Author's response to reviews

Title: What zinc supplementation does and does not achieve in diarrhea prevention: A systematic review and meta-analysis

Authors:

Archana B Patel (dr_apatel@yahoo.com)
Manju Mamtani (manjuhemant@gmail.com)
Neetu Badhoniya (neets_badoniya@yahoo.com)
Hemant Kulkarni (hemant_kulkarnius@yahoo.com)

Version: 4 Date: 8 May 2011

Author's response to reviews:

Thanks for your note. We have now done two additional changes - first we have ensured that the word count for each figure is less than 250; and second we have reformatted the author contribution section as required. Hope this helps.

hemant